Author's response to reviews

Title: Localization of uPAR and MMP-9 in lipid rafts is critical for migration, invasion and angiogenesis in human breast cancer cells.

Authors:

Hari Raghu (hraghu@uic.edu)
Prasanna Kumar Sodadasu (drprasanna1611@gmail.com)
Rama Rao Malla (rrmalla@uic.edu)
Christopher S. Gondi (gondi@uic.edu)
Norman Estes (nestes@uic.edu)
Jasti S. Rao (jsrao@uic.edu)

Version: 3 Date: 23 November 2010

Author's response to reviews:

November 23, 2010

The Editor
BMC Cancer

RE: 2nd Revision of MS: 6444493314151827 – Localization of uPAR and MMP-9 in lipid rafts is critical for migration, invasion and angiogenesis in human breast cancer cells. - Reg

Dear Editor:

As suggested in your email dated Nov.22nd 2010, regarding the above mentioned manuscript submitted to BMC Cancer, we have made the necessary corrections mentioned below and have uploaded the revised manuscript in your website.

Major revisions

Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution

As mentioned, we have made the necessary changes to the affiliations of every author as requested by the editorial team.

Figures: Please reduce the volume of text presented each figure legend to no more than 300 words or it will compete with the image for space in the final publication.

As suggested, the volume of the text presented in each figure legend has been
reduced to less than 300 words.

Minor revisions

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

As suggested highlighting and tracking has been removed from the revised manuscript.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

The manuscript has been checked for typographical errors and corrections have been made accordingly in the revised manuscript.

I hope the necessary changes to the manuscript are as requested by the editorial team. If you have any questions or require additional information, please feel free to contact me.

Sincerely,

Jasti S. Rao, Ph.D.
Department Head and Program Director